Author's response to reviews

Title: Magnetic resonance imaging after most common form of concussion

Authors:

Harald Schrader (harald.schrader@ntnu.no)
Dalia Mickeviciene (daliamickeviciene@gmail.com)
Rymante Gleizniene (rymangleiz@gmail.com)
Sylvija Jakstiene (sylvijakstiene@zebra.lt)
Danguole Surkiene (surkiene@gmail.com)
Lars J Stovner (lars.stovner@ntnu.no)
Diana Obelieniene (dianobel@takas.lt)

Version: 4 Date: 13 June 2009

Author's response to reviews:

Only major and minor revisions as required by the editor have been made (except for omission of the "s" in conclusions)